Author's response to reviews

Title: Novel Deletion Alleles Carrying CYP21A1P/A2 Chimeric Genes In Brazilian Patients With 21-Hydroxylase Deficiency

Authors:

Fernanda B Coeli (fbcoeli@bol.com.br)
Fernanda C Soardi (soardi@unicamp.br)
Renan D Bernardi (renandarin@gmail.com)
Marcela de Araújo (maraujo@boldrini.org.br)
Luciana C Paulino (luciana.paulino@ufabc.edu.br)
Ivy F Lau (mpmello276@hotmail.com)
Reginaldo J Petroli (rpetroli@unicamp.br)
Sofia HV Lemos-Marini (sofia.lemos@terra.com.br)
Maria TM Baptista (terma@uol.com.br)
Gil Guerra Jr (gileandrea@uol.com.br)
Maricilda Palandi de Mello (mmello@unicamp.br)

Version: 4 Date: 28 May 2010

Author's response to reviews: see over
May 28th, 2010.

To
Scott Edmunds PhD
The BioMed Central Editorial Team
e-mail: editorial@biomedcentral.com
Tel: +44 (0) 20 3192 2013

MS: 1002574954320354 Formatting changes

Dear Dr. Edmunds,

This is to submit the formatted manuscript entitled “Novel Deletion Alleles Carrying CYP21A1P/A2 Chimeric Genes In Brazilian Patients With 21-Hydroxylase Deficiency” to be published in BMC Medical Genetics.

The manuscript has been revised. We have addressed major requests as follow:

**Major revisions**

Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is “First name Initial Last name” (e.g. Joe F Bloggs).

We have changed the style author list, however, for one of the author, the name in the submission form is not the corrected (Gil Guerra Jr) the correct form for this author is Gil Guerra-Júnior.

References: Please make sure references are cited in numerical order. We notice reference 30 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.

Reference 30 is on the text.

References: Please make sure all references are cited in numerical order. Some of your reference citations are out of sequence.

Reference sequence was corrected.

Table 3 unsuitable: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.
We have modified table 3. It has been created with table tool. As it is very large, we inverted rows to columns and columns to rows. We also eliminated last four rows and included those data as table footnotes.

Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend, preferably indicated with bold text.

We have changed figure titles.

Manuscript: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Authors’ information (optional); Acknowledgements (optional); References

We have changed for the correct structure.

Minor revisions

Title page: Please remove running title and key words, and move disclosure statement to the end of the manuscript, just before the references. Please also rename it 'Competing Interests'.

We have changed.

Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is First name Initial Last name (e.g. Joe F Bloggs).

We have changed the style author list, however, for one of the author, the name in the submission form is not the corrected (Gil Guerra Jr) the correct form for this author is Gil Guerra-Júnior.

Figures: We note that the figures have been included in the manuscript file. Please upload the figures as separate figure files using the upload form on the submission system only, and delete the figures from the manuscript file. The figure file should not include the title (e.g. Figure 1... etc.) or the figure number. The figures are numbered automatically in the order in which they are uploaded.

We have done.
Figures: Please remove the figure names/numbers from the image files.

We have done.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

We have done.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have checked.

We thank you for your attention.

Yours sincerely,

Maricilda Palandi de Mello, Ph.D.